Author's response to reviews

Title: Prevalence of obesity in a rural Asian Indian (Bangladeshi) population and its determinants

Authors:

Tasnima Siddiquee Dr (tasnimasiddiquee08@yahoo.com)
Bishwajit Bhowmik Dr (doctorbiplob@gmail.com)
Nayla CD V Moreira Dr (n.c.d.v.moreira@studmed.uio.no)
Hajera Mahtab Prof (hajeramahtab@gmail.com)
AK A Khan Prof (president@dab-bd.org)
Akhtar Hussain Prof (hussain.akhtar@medisin.uio.no)

Version: 2 Date: 5 August 2014

Author's response to reviews:

Dear Sir
We have added the line number as per your instruction.
Yours sincerely,
Dr. Bishwajit Bhowmik